Author's response to reviews

Title: Prevalence of Trichomonas vaginalis Infection among Egyptian Women using Culture and Latex agglutination: Cross-Sectional Study

Authors:

Ahmed Mahmoud MD (ahmad.mahmoud.ali@hotmail.com)
Nadine A Sherif MD (nadine.sherif@kasralainy.edu.eg)
Rana Abdella MD (rana.abdella@yahoo.com)
Amira R El-Genedy MBBCh (amiraraafat@live.com)
Abdalla Y El Kateb MD (aelkateb@hotmail.com)
Ahmed NH Askalani MD (a_askalani@hotmail.com)

Version:8 Date:22 November 2014

Author's response to reviews: see over
Reviewer's report

Title:
Prevalence of Trichomonas Vaginalis Infection in Egyptian Women: Cross-Sectional Study

Version: 7
Date: 28 October 2014
Reviewer: Barbara Body

Reviewer's report:

Major compulsory revisions:
I have no additional suggestions for compulsory revisions. I am grateful to the editors and authors for their patience with my concerns in this review process.

Minor essential revisions
There are still typographical errors but most will be addressed in the galley process. All the manuscript has been edited and revised again.

Level of interest: An article whose findings are important to those with closely related research interests

Quality of written English: Fine

Statistical review: Yes, and I have assessed the statistics in my report.

Declaration of competing interests: I have participated in clinical trials for the FDA clearance of nucleic acid amplification tests from two manufacturers. I did not receive compensation but the company for which I work was paid to perform the testing. I do not have any other financial or non-financial competing interests

Level of interest:
An article whose findings are important to those with closely related research interests

Quality of written English: Acceptable

Statistical review: Yes, and I have assessed the statistics in my report.
Declaration of competing interests: I have participated in clinical trials for the FDA clearance of nucleic acid amplification tests from two manufacturers. I did not receive compensation but the company for which I work was paid to perform the testing. I do not have any other financial or non-financial competing interests.

Editorial Requirements:

Post-review copyediting: We recommend that you copyedit the paper to improve the style of written English. If this is not possible, you may need to use a professional language editing service. The Manuscript has been copyedited by a native English speaker with a medical background.